

University of Kentucky
UKnowledge

Sanders-Brown Center on Aging Faculty Publications

Aging

2-3-2021

### Novel Influences of Sex and *APOE* Genotype on Spinal Plasticity and Recovery of Function after Spinal Cord Injury

Lydia E. Strattan University of Kentucky, hager1252@uky.edu

Daimen R. Britsch University of Kentucky, drsb25@uky.edu

Chris M. Calulot University of Kentucky, Chris.Calulot@uky.edu

Rachel S. J. Maggard University of Kentucky, rachel.maggard@uky.edu

Erin L. Abner University of Kentucky, erin.abner@uky.edu

See next page for additional authors Follow this and additional works at: https://uknowledge.uky.edu/sbcoa\_facpub

Part of the Geriatrics Commons, Medical Physiology Commons, and the Neurosciences Commons Right click to open a feedback form in a new tab to let us know how this document benefits you.

### **Repository Citation**

Strattan, Lydia E.; Britsch, Daimen R.; Calulot, Chris M.; Maggard, Rachel S. J.; Abner, Erin L.; Johnson, Lance A.; and Alilain, Warren J., "Novel Influences of Sex and *APOE* Genotype on Spinal Plasticity and Recovery of Function after Spinal Cord Injury" (2021). *Sanders-Brown Center on Aging Faculty Publications*. 162.

https://uknowledge.uky.edu/sbcoa\_facpub/162

This Article is brought to you for free and open access by the Aging at UKnowledge. It has been accepted for inclusion in Sanders-Brown Center on Aging Faculty Publications by an authorized administrator of UKnowledge. For more information, please contact UKnowledge@lsv.uky.edu.

### Novel Influences of Sex and *APOE* Genotype on Spinal Plasticity and Recovery of Function after Spinal Cord Injury

Digital Object Identifier (DOI) https://doi.org/10.1523/eneuro.0464-20.2021

### **Notes/Citation Information**

Published in *eNeuro*.

Copyright © 2021 Strattan et al.

This is an open-access article distributed under the terms of the Creative Commons Attribution 4.0 International license, which permits unrestricted use, distribution and reproduction in any medium provided that the original work is properly attributed.

### Authors

Lydia E. Strattan, Daimen R. Britsch, Chris M. Calulot, Rachel S. J. Maggard, Erin L. Abner, Lance A. Johnson, and Warren J. Alilain



Research Article: New Research | Disorders of the Nervous System

### Novel influences of sex and APOE genotype on spinal plasticity and recovery of function after spinal cord injury

https://doi.org/10.1523/ENEURO.0464-20.2021

Cite as: eNeuro 2021; 10.1523/ENEURO.0464-20.2021

Received: 28 October 2020 Revised: 30 December 2020 Accepted: 22 January 2021

This Early Release article has been peer-reviewed and accepted, but has not been through the composition and copyediting processes. The final version may differ slightly in style or formatting and will contain links to any extended data.

Alerts: Sign up at www.eneuro.org/alerts to receive customized email alerts when the fully formatted version of this article is published.

Copyright © 2021 Strattan et al.

This is an open-access article distributed under the terms of the Creative Commons Attribution 4.0 International license, which permits unrestricted use, distribution and reproduction in any medium provided that the original work is properly attributed.

- 1 Title: Novel influences of sex and *APOE* genotype on spinal plasticity and recovery of function
  - 2 after spinal cord injury
  - 3 Abbreviated title: Sex and APOE genotype impact spinal plasticity after SCI
  - 4 Lydia E. Strattan<sup>1</sup>, Daimen R. Britsch<sup>2</sup>, Chris M. Calulot<sup>2</sup>, Rachel S. J. Maggard<sup>1</sup>, Erin L.
  - 5 Abner<sup>3,4,5</sup>, Lance A. Johnson<sup>3,6</sup>, Warren J. Alilain<sup>1,2</sup>
  - <sup>1</sup>Department of Neuroscience, University of Kentucky College of Medicine, Lexington, KY,
     United States
  - <sup>2</sup>Spinal Cord and Brain Injury Research Center, University of Kentucky College of Medicine,
     Lexington, KY, United States
- 10 <sup>3</sup>Sanders-Brown Center on Aging, University of Kentucky, Lexington, KY, United States
- <sup>4</sup>Department of Biostatistics, University of Kentucky, Lexington, KY, United States
- 12 <sup>5</sup>Department of Epidemiology, University of Kentucky, Lexington, KY, United States
- <sup>6</sup>Department of Physiology, University of Kentucky College of Medicine, Lexington, KY,
   United States
- 15 Author Contributions: LS, EA, LJ, CC, and WA designed research. LS, DB, RM, and CC
- 16 performed research. LS, EA, and WA analyzed data. LJ contributed reagents/analytical tools. LS,
- 17 EA, and WA wrote the paper.
- 18 Corresponding author: Warren J. Alilain, PhD, Email: warren.alilain@uky.edu
- 19 Number of Figures: 7
- 20 Number of Tables: 1
- 21 Number of Multimedia: 0
- 22 Number of words in Abstract: 245
- 23 Number of words for Significance Statement: 112
- 24 Number of words in Introduction: 732
- 25 Number of words in Discussion: 2133
- 26 Acknowledgements: We thank Kyle Ritter, B.S., Emily Huffman, B.S., Jessica Newton, M.S.,
- 27 and Laura Mendenhall for their assistance in completing experiments. We also thank Matt
- 28 Hazzard at the University of Kentucky College of Medicine for designing circuitry illustrations.

- 30 Conflicts of interest: Authors report no conflict of interest.
- 31
- 32 Funding Sources: This material is based upon work supported by the National Science
- 33 Foundation Graduate Research Fellowship under Grant No. 1839289. This work was also
- 34 supported by the Craig H. Neilsen Foundation (598741), and the NIH R01(NS101105).
- 35

### 36 Abstract

37 Spinal cord injuries can abolish both motor and sensory function throughout the body. 38 Spontaneous recovery after injury is limited and can vary substantially between individuals. 39 Despite an abundance of therapeutic approaches that have shown promise in preclinical models, 40 there is currently a lack of effective treatment strategies that have been translated to restore 41 function after SCI in the human population. We hypothesized that sex and genetic background of injured individuals could impact how they respond to treatment strategies, presenting a barrier to 42 43 translating therapies that are not tailored to the individual. One gene of particular interest is 44 APOE, which has been extensively studied in the brain due to its allele-specific influences on 45 synaptic plasticity, metabolism, inflammation, and neurodegeneration. Despite its prominence as 46 a therapeutic target in brain injury and disease, little is known about how it influences neural 47 plasticity and repair processes in the spinal cord. Utilizing humanized mice, we examined how 48 the  $\varepsilon_3$  and  $\varepsilon_4$  alleles of *APOE* influence the efficacy of therapeutic intermittent hypoxia (IH) in 49 inducing spinally-mediated plasticity after cervical SCI. IH is sufficient to enhance plasticity and restore motor function after experimental SCI in genetically similar rodent populations, but its 50 effect in human subjects is more variable (Golder, 2005; Hayes et al., 2014). Our results 51 52 demonstrate that both sex and APOE genotype determine the extent of respiratory motor 53 plasticity that is elicited by IH, highlighting the importance of considering these clinically 54 relevant variables when translating therapeutic approaches for the SCI community.

### 56 Significance Statement

57	There is currently a critical need for therapeutics that restore motor and sensory function
58	effectively after cervical spinal cord injury. Although many therapeutic approaches, including
59	intermittent hypoxia, are being investigated for their potential to enhance spinal plasticity and
60	improve motor outcomes after SCI, it is unknown whether the efficacy of these treatment
61	strategies is influenced by individuals' genetic background. Here we show that APOE genotype
62	and sex both play a role in determining the propensity for motor plasticity in humanized mice
63	after cervical SCI. These results indicate that sex and genetic background dictate how individuals
64	respond to therapeutic approaches, thereby emphasizing the importance of developing
65	personalized medicine for the diverse SCI population.

### 67 Introduction

68 Over 17,000 Americans experience a spinal cord injury every year (National Spinal Cord Injury Statistical Center, 2018). Depending on the level of injury, damage to neural pathways in 69 70 the spinal cord can lead to a multitude of sensory deficits and loss of crucial motor functions. 71 Over the past few decades, many promising therapeutic approaches have been developed to 72 enhance neuroprotection or induce anatomical and functional plasticity of spinal pathways to 73 restore function (David D. Fuller et al., 2003; Huie et al., 2017; Satkunendrarajah et al., 2018; 74 Zholudeva et al., 2018; Jack et al., 2020). Moreover, pivotal studies using nerve grafts, PTEN 75 deletion, NOGO inhibition, or degradation of the perineuronal net or chondroitin sufate 76 proteoglycans (CSPGs) have demonstrated that the CNS is capable of overcoming neural 77 intrinsic and extrinsic barriers to regeneration after injury, leading to meaningful preclinical 78 recovery (David & Aguayo, 1981; Chen et al., 2000; Park et al., 2008; Alilain et al., 2011; Urban et al., 2019). However, these therapeutic strategies have met with varied clinical success and 79 80 there remains a lack of effective treatment strategies for the human SCI population (reviewed by 81 Ahuja et al., 2017).

82 A striking difference between individuals living with SCI and the animals used to model them is 83 the level of genetic diversity represented in these populations. In contrast to the incredible 84 diversity of the human population, preclinical studies typically utilize homogenous groups of 85 animals with the same sex and similar genetic backgrounds. While this does facilitate easier 86 determination of treatment effects, it also makes it less likely that discoveries in these models 87 will translate to human patients. Although an increasing number of preclinical investigations are 88 addressing how sex influences the efficacy of therapeutic strategies, the impact of genetic 89 variability remains largely unexplored. A recent review by Fouad et al. specifically outlined the

90 importance of evaluating the influence of factors such as sex and genotype to address the

91 neuroanatomical-functional paradox and lack of therapeutic translation in SCI (Fouad et al.,

92 2020). Indeed, we hypothesize that genetic factors could play a considerable role in determining

93 how individuals respond to treatment strategies.

- 94 Apolipoprotein E (ApoE) is a highly expressed lipid carrier in the CNS (Boyles et al., 1985). It is 95 encoded by the APOE gene, which exists in three common alleles designated  $\varepsilon_2$ ,  $\varepsilon_3$ , and  $\varepsilon_4$ . The 96  $\epsilon$ 4 allele, which is carried by nearly 1 in 5 individuals, has been associated with a number of 97 detrimental outcomes, including a weakening of synaptic plasticity in the brain (Zhao et al., 98 2018). However, despite a robust body of literature in neurodegenerative diseases and traumatic 99 brain injury (Zhou et al., 2008; Mahley, 2016; Main et al., 2018), the impact of  $\varepsilon$ 4 on plasticity in 100 the spinal cord remains underexplored. We hypothesized that spinally-mediated plasticity is 101 constrained in apoE4 animals, thereby demonstrating the importance of considering the diversity 102 of the human population when developing therapeutic approaches for people with SCI. 103 To test this hypothesis, we utilize a model of cervical injury to examine how APOE genotype 104 alters the response to intermittent hypoxia (IH). Most SCIs occur at these high levels and can 105 disrupt the neural circuitry that mediates breathing, leading to respiratory insufficiency and 106 potentiating the need for mechanical ventilation (Bergofsky, 1964; Alp & Voss, 2006; National 107 Spinal Cord Injury Statistical Center, 2018). Mechanical ventilation increases the risk of 108 respiratory infection, a leading cause of rehospitalization and death following cervical spinal
- 109 cord injury (cSCI) (DeVivo & Ivie, 1995).
- 110 In recent years, there has been a growing appreciation for the potential of intermittent hypoxia
- 111 (IH) as a treatment strategy for a host of conditions including SCI (Navarrete-Opazo & Mitchell,

112 2014). In clinical trials, therapeutic IH has been utilized to increase limb function and to 113 facilitate ventilation in persons with SCI by enhancing plasticity in the spinal cord (Tester et al., 114 2014; Lynch et al., 2017; Trumbower et al., 2017). Neural pathways in the cervical region which 115 mediate breathing are critical therapeutic targets of IH, including spared pathways which might 116 remain after injury. However, the influence of human genetic variability on IH-induced recovery 117 is unknown. Therefore, we utilized this model of spinally-mediated plasticity to examine how 118 expression of different human APOE alleles alter the efficacy of therapeutic strategies, such as 119 IH, that are being developed to enhance plasticity following SCI. Our results provide evidence 120 that both sex and APOE genotype determine the propensity for plasticity in humanized mice that 121 are exposed to therapeutic IH.

### 122 Materials and Methods

### 123 C2 Hemisections

124 All experiments were approved by the Institutional Animal Care and Use Committee at 125 the University of Kentucky. Mice expressing human APOE isoforms under control of the mouse 126 APOE promotor (targeted replacement mice) were backcrossed for at least 10 generations to the 127 C57BL/6 background (Sullivan et al., 1997; Sullivan et al., 1998; Knouff et al., 1999). Mice 128 were group-housed on a twelve-hour light/dark cycle and fed normal chow diet ad libitum. All 129 mice were 92-105 days old at the time of injury. Female (20-24g) and male (22-30g) mice were 130 anesthetized with isoflurane. Animals were then prepped for surgery by shaving the surgical area 131 followed by disinfecting with alternating betadine and 70% ethanol swabs. Puralube ophthalmic 132 ointment was applied to the eves to prevent drying during surgery. A midline incision was made 133 through the skin just caudal to the ears to between the scapulae. Marcaine/bupivacaine was 134 instilled along the incision site. The acromiotrapezius, semispinalis capitus, and rectus capitus

135	posterior muscles were cut along their midline, bluntly dissected, and retracted. Paravertebral
136	muscles were cut away from the C2 vertebra using ToughCut Spring Scissors (Fine Science
137	Tools). The lamina of the C2 vertebra was then removed using Spring Scissors. Under a
138	dissecting microscope (Meiji EMZ), a left lateral C2 hemisection (C2Hx) was performed by
139	inserting a 27-gauge needle into the midline of the spinal cord at the C2 level and dragging the
140	needle to the left lateral edge of the cord. This was then repeated once to ensure a complete
141	injury. Musculature was sutured (6-0 absorbable suture) and skin was closed with Vetbond
142	Tissue Adhesive (3M). Animals received subcutaneous buprenorphine (0.75mg/kg) and
143	carprofen (10mg/kg) immediately after surgery. Male mice were housed individually following
144	surgery to prevent fighting amongst cagemates.

145 Intermittent Hypoxia and Diaphragmatic Electromyography (EMG)

146 Three weeks after hemisection, animals were anesthetized with isoflurane using the 147 SomnoSuite Anesthesia System (Kent Scientific). A laparotomy was performed by cutting 148 through the rectus abdominis, external oblique, and internal oblique muscles. Bipolar electrodes, 149 connected to an amplifier and data acquisition system (CWE BMA-400 Four-channel 150 Bioamplifier, CED 1401 with Spike2 Data Analysis Computer Interface), were inserted into the 151 dorsal region of the left hemidiaphragm, where they were secured using Vetbond. Bilateral 152 recordings were not performed due to the increased attrition rate we observed after performing 153 bilateral electrode insertion. The laparotomy was also closed using Vetbond. Ten minutes of 154 baseline breathing activity was recorded. The air input to the Somnosuite was then changed from 155 room air (normoxia) to a tank of 11% oxygen, 89% nitrogen gas (hypoxia) for 5 min, at which 156 point it was switched back to room air for 5 minutes. This was repeated for 3 bouts of hypoxia 157 separated by 5 minutes of normoxia. Diaphragmatic activity was recorded for 1 hour after the

158 final hypoxic bout (Fig. 1 *A*). Although core temperature was not monitored during recordings,
159 animals were kept on heating pads throughout all recording procedures to maintain body
160 temperature.

### 161 Sectioning and Staining

To harvest tissue for immunohistochemistry, mice were perfused with 4% paraformaldehyde
(PFA) following the diaphragmatic EMG recording. The spinal column was isolated and placed
in 4% PFA at 4°C. After two days, tissue was removed from PFA and placed in a 30% sucrose
solution at 4°C for cryoprotection until sectioning.

166 Tissue was mounted and frozen in Tissue Plus O.C.T. Compount (Fisher Healthcare) and 167 cut at a thickness of 20µm on a cryostat (Leica). Serial sections from the injury site (C1-C2) 168 were placed on one set of slides while serial sections from the level of the PMN (C3-C6) were 169 placed on another set. Injured tissue slides were dehydrated in ethanol and stained with 0.1% 170 Cresyl violet solution (Sigma Cat #C5042). Slides were then mounted using permount (Electron 171 Microscopy Sciences Cat #17986-01). For 5-HT staining, frozen section were thawed to room 172 temperature, rinsed with 1x PBS, and blocked in a solution of 5% normal goat serum, 0.1% bovine serum albumin, and 0.1% TritonX-100 dissolved in PBS. Slides were incubated in 5-HT 173 174 primary antibody diluted 1:10,000 (rabbit, ImmunoStar Cat #20080) then goat anti-rabbit 175 AlexaFluor488 secondary antibody (1:500, Life Technologies Cat #A11034). Stained slides were 176 mounted with ProLong Gold mountant with DAPI (Invitrogen Cat #P36931). For WFA 177 (Wisteria Floribunda Lectin) staining, frozen sections were thawed to room temperature, washed 178 with 1xPBS, then blocked in 3% NGS diluted in PBS. Slides were then incubated in WFA

- 179 primary antibody conjugated to Fluorescein at a dilution of 1:400 (Vector Labs Cat #FL-1351).
- 180 Stained slides were mounted in ProLong Gold with DAPI.

181 *Trace Analysis* 

After recording, raw diaphragmatic EMG was rectified and integrated using Spike2 software. Analysis was performed at twelve time points: twice during baseline recording, once during each hypoxic and normoxic bout, and at 10, 20, 30, and 40 minutes after the final hypoxic period (Figure 1 *A*). For each time point, peak amplitude was averaged over a 30 second period. Amplitude of diaphragmatic bursts at each time point were normalized to that animal's prehypoxia baseline amplitude. Frequency of diaphragmatic bursts, indicative of breaths, was also quantified over a 30 second period at each time point.

189 Imaging and image quantification

190 Staining for cresyl violet and WFA was imaged on a Keyence BZ-X810 fluorescence 191 microscope for quantification. Cresyl violet stained sections were imaged using brightfield 192 illumination at 2x. Sections stained for WFA were imaged at 10x using the monochromatic 193 camera with high resolution (0.75488µm/pixel) for quantification. Additional images for 194 publication were acquired on a Nikon Eclipse Ti series inverted confocal at 40x, focused on the 195 ventral horn in the region of the putative PMN. Sections stained for 5-HT were imaged on the 196 Nikon at 20x. Images of 5-HT staining for publication were taken at 40x in the region of the 197 PMN. All imaging and quantification was performed on the ventral horn of the left side of the 198 cord, ipsilateral to the injury.

WFA labeling was quantified using the HALO image analysis platform (Indica Labs).We developed and optimized an algorithm on the Area Quantification v1.0 to capture positive

201	staining for WFA while omitting any nonspecific fluorescence. A region of interest (ROI) was
202	drawn around the left ventral horn of sections at the level of C4. The quantification algorithm
203	was applied to the ROI of each section. The area of staining was then normalized to the total area
204	of the ROI. Three tissue sections at level C4 were analyzed for each animal. 5-HT labeling was
205	also quantified with HALO. A region of interest was drawn around the left ventral horn.
206	Serotonergic fibers within the ROI were traced using the embedded annotation tool. The total
207	length of fibers was then normalized to the area of the ROA.
208	Experimental Design and Statistical Analyses
209	Sample sizes for mice receiving diaphragmatic EMG recordings were calculated based on
210	preliminary data from 10 hemisected mice representing all three genotypes using Cohen's D to
211	measure effect size. Group sizes for each sex and genotype are found in Table 1. Tissue from a

subset of animals was perfused with PFA and spinal cord tissue was harvested from these

213 animals for IHC and quantification of WFA and serotonergic sprouting (apoE3 n=4, apoE4 n=5)

214 For statistical analysis of EMG traces, repeated measures (RM)-ANOVA was used to account for

215 within-subject correlation given repeated measurements over time. Stratified RMANOVA

analyses were performed on male and female traces. Results were considered statistically

217 significant if t≥1.96. Student's t-test was used to analyze 5-HT fiber staining on spinal cord

218 tissue. Welch's t-test for unequal variances was used to analyze staining of WFA. Results were

219 considered statistically significant if p < 0.05. Investigators were blinded until all diaphragmatic

EMG and histology analyses were complete. The mean difference (MD) and 95% confidence

221 interval (CI) were calculated to provide an estimate of the range of possible differences between

222 groups.

Data Structure	Type of Test	95% confidence interval
<i>a.</i> Repeated measurements, normal distribution	RMANOVA	-0.2202 to 0.3063
<i>b.</i> Normal distribution with within-subject correlation	Paired t-test	-0.1570 to 0.2160
<i>c.</i> Normal distribution with within-subject correlation	Paired t-test	-0.2907 to 0.1235
<i>d.</i> Repeated measurements, normal distribution	RMANOVA	-0.2958 to 0.35575
<i>e</i> . Repeated measurements, normal distribution	RMANOVA	-1.17798 to 0.07798
<i>f.</i> Repeated measurements, normal distribution	RMANOVA	-1.53798 to -0.2820
g. Repeated measurements, normal distribution	RMANOVA	-0.9958 to -0.3442
<i>h</i> . Repeated measurements, normal distribution	RMANOVA	-0.75238 to 0.11238
<i>i.</i> Repeated measurements, normal distribution	RMANOVA	0.426832 to 1.25317
<i>j.</i> Repeated measurements, normal distribution	RMANOVA	1.66683 to 2.49317
<i>k.</i> Repeated measurements, normal distribution	RMANOVA	0.64132 to 1.37868
<i>l.</i> Normal distribution, Unequal variance	Welch's t-test	-0.0003607 to 0.006133
<i>m</i> . Normal distribution	Student's t-test	0.0003556 to 0.002912
<i>n</i> . Normal distribution	Student's t-test	-0.001091 to 0.003178
<i>o</i> . Normal distribution	Student's t-test	-0.2916 to 1.237

223

### 224 Results

225 Respiratory motor plasticity in C2 hemisected humanized APOE mice

At 3 months of age, male and female mice received a left C2 hemisection by making an

227 incision from the midline to the left lateral edge of the spinal cord just caudal to the C2 dorsal

228 roots. This injury effectively disrupts the neural circuitry that descends from the ipsilateral

229	medullary respiratory nuclei to phrenic motor neurons on the left side (Fig. 1 B). At the time of
230	injury, hemisection was visually confirmed by observing the thorax of each mouse to ensure that
231	only the right side of the thorax continued rhythmically expanding with each breath. Injury
232	completeness was histologically confirmed upon sacrifice of a subset of mice (n=16) using cresyl
233	violet (Fig. 1 C). All mice were homozygous for human $\varepsilon$ 3 or $\varepsilon$ 4 alleles expressed under control
234	of the murine APOE promoter as described previously (Patrick M. Sullivan et al., 1997; Knouff
235	et al., 1999). At 3 weeks post-injury, mice were exposed to intermittent hypoxia. This consisted
236	of 3 hypoxic (11% $O_2$ ) bouts of 5 minute duration separated by 5 minutes of normoxia as
237	illustrated in Figure 1. We evaluated the breathing response to IH by concurrently recording
238	diaphragmatic EMG, which continued for 1 hour following the final hypoxic bout. Amplitude of
239	diaphragmatic bursts was quantified while blinded and then grouped according to APOE
240	genotype. No difference was found in the response to IH between mice expressing $\epsilon 3$ or $\epsilon 4$ (Fig.
241	1 D, RMANOVA p=0.741). All animals appear to experience an initial decrease in
242	diaphragmatic activity during the first hypoxic bout. Breathing in both apoE3 and apoE4 mice
243	remained constant once the IH protocol ended (Fig. 1 D).
244	Previous studies in rodents (Bach & Mitchell, 1996; Baker & Mitchell, 2000; reviewed
245	by Fuller et al., 2000; Terada et al., 2008)) have shown that IH treatment gives rise to an
246	augmentation of breathing activity that characterizes LTF. We therefore compared amplitude at
247	the beginning of IH and 40 minutes after the final bout of hypoxia to determine whether
248	breathing activity increased in response to IH. Neither genotype exhibited significant
249	augmentation of diaphragmatic activity at 40 minutes post-hypoxia, indicative of a lack of LTF
250	in the humanized mice (Fig. 1 <i>E</i> , paired t-test E3 p=0.741, E4 p=0.405).

251 Sex differences in apoE modulation of LTF

	252	To inve
	253	from males and
	254	days after inju
Ц	255	When compari
Ö	256	genotypes in m
_	257	representative
$\mathbf{O}$	258	genotypes exhi
	259	significant gen
	260	RMANOVA, J
ס	261	spontaneous re
$\geq$	262	contrast, the ov
	263	spontaneous fu
a)	264	and females fro
Ţ	265	spontaneous re
	266	EMG data dem
Ű	267	diaphragmatic
Ŭ	268	recording perio
$\triangleleft$	269	significance (R
0	270	from apoE4, w
<u> </u>	271	RMANOVA, t
0	272	Analys
~	273	negative trend
Ð	274	- RMANOVA, p
		-

-	To investigate sex-dependent influences of <i>APOE</i> genotype on LTF, we separated data
3	from males and females for independent analysis. Animals were weighed every day for the first 4
ŀ	days after injury and then once a week until diaphragmatic EMG recordings were performed.
5	When comparing weights over time after injury, there was no significant difference between
6	genotypes in male (p=0.16) or female (p=0.65) mice (data not shown). Figure 2 A shows
,	representative traces from male apoE3 and E4 mice. As evidenced in these traces, both
3	genotypes exhibited a decrease in frequency over time after IH. However, there was no
)	significant genotype effect on the magnitude of this decrease (Fig. 2 A, extended Fig. 2-1 A,
)	RMANOVA, p=0.846). Previous studies in rats (Warren et al., 2018) have reported no
	spontaneous recovery in the paralyzed hemidiaphragm even chronically after C2 hemisection. In
2	contrast, the overwhelming majority of the 32 mice used in the current study showed
3	spontaneous functional recovery of the paralyzed mouse hemidiaphragm. Considering all males
ŀ	and females from which we recorded diaphragmatic EMG's, only 2 mice displayed no
5	spontaneous recovery: 1 male of each genotype (Fig 2 $B$ ). Quantification of the diaphragmatic
6	EMG data demonstrates that males expressing the $\varepsilon 3$ allele display a decline in the amplitude of
,	diaphragmatic bursting beginning in the first hypoxic bout and persisting throughout the
3	recording period (Fig. 2 C). Although this deterioration of activity did not reach statistical
)	significance (RMANOVA, t=0.03), it is worthwhile to highlight how the apoE3 males diverged
)	from apoE4, which demonstrate slightly heightened activity at 40 minutes post-IH (Fig. 2 C,D,
	RMANOVA, t=-0.65).
2	Analysis of diaphragmatic EMGs in female mice of both genotypes showed a similar

negative trend in the breathing frequency induced by IH (Fig. 3 *A*, extended Fig. 3-1,

RMANOVA, p=0.673). A subset of mice displayed a complete loss of diaphragmatic activity

275	following hypoxic exposure. We refer to these animals as "non-responders". Three non-
276	responders emerged in the apoE4 group, while none were observed in the mice that expressed $\epsilon 3$
277	(representative trace shown in Fig. 3 <i>B</i> ). However, unlike the male mice, all females
278	demonstrated spontaneous recovery prior to IH (data not shown). Quantification of
279	diaphragmatic burst amplitude in females that maintained diaphragmatic activity after IH showed
280	that apoE3 mice responded to IH with an initial decrease in burst amplitude. This decline was
281	temporary and activity returned to near baseline levels by 40 minutes (Fig. 3 C). However,
282	apoE4 females exhibited an immediate reduction in burst amplitude that is still evident the end of
283	the recording period. At 40 minutes post-IH, breathing of apoE4 females is significantly
284	depressed compared to that of apoE3's at the same time point (Fig. 3 C,D, mixed model
285	RMANOVA t=2.08). Consistent with the combined data, none of the female mice expressing
286	human APOE developed the gradual and prolonged breathing augmentation that is characteristic
287	of LTF.

288 When animals are challenged with a brief bout of hypoxia, feedback from peripheral 289 chemoreceptors induces an augmentation of respiratory output. This change in ventilation is 290 known as the hypoxic ventilatory response (HVR, described by Pamenter & Powell, 2016). 291 During the hypoxic bouts of IH treatment, all apoE mice exhibited a decline in diaphragmatic 292 activity instead of the expected amplification. To further investigate the HVR in our humanized 293 mice, we exposed an additional, smaller cohort of mice to a 10-minute bout of hypoxia and 294 assessed the changes in amplitude and frequency of diaphragmatic bursting. No females of either 295 genotype displayed an increase or decrease in amplitude, but breath frequency began to decline 296 by the end of the hypoxic period in those expressing  $\varepsilon 4$  (extended Fig. 3-1 B,C). In male mice 297 expressing  $\varepsilon 3$ , there was a sharp decline in both amplitude and frequency of diaphragmatic firing in response to hypoxia such that breathing activity was abolished at 10 minutes. Conversely,
amplitude and frequency in apoE4 males remained constant during hypoxia (extended Fig. 2-1 *B*,*C*).

### 301 Perineuronal net upregulation and serotonergic sprouting in the phrenic motor nucleus

302 Secretion of chondroitin sulfate proteoglycans (CSPGs) is upregulated after SCI in wild 303 type animals, creating a barrier to plasticity, regeneration, and sprouting (Tom et al., 2009; 304 Alilain et al., 2011). Thus far, it is unknown if the magnitude of this upregulation is modulated 305 by human APOE genotype. Therefore, we utilized WFA staining to compare the amount of PNN 306 present in injured spinal cords at the C4 level to determine if the IH-induced reduction in 307 diaphragmatic activity observed in E4 females was correlated with increased amounts of 308 inhibitory PNN. Indeed, we found that apoE4 females tended to have a higher density of WFA 309 around the phrenic motor nucleus after injury, although this trend did not reach significance (Fig 310 4 A-C, Welch's t-test, p=0.0697).

311 The PNN can limit 5-HT sprouting after injury (Alilain et al., 2011). To determine 312 whether differences in respiratory motor plasticity observed in females were due to the amount 313 of serotonin at the level of the phrenic motor nucleus after C2Hx, serotonergic fibers were 314 labeled and quantified in the ventral horn ipsilateral to injury. Serotonergic sprouting after injury 315 has previously been correlated with the restoration of breathing function and enhancement of 316 LTF (Golder, 2005). We postulated that dampened respiratory plasticity in  $\varepsilon 4$  females may be due to a lack of serotonergic sprouting after injury. Surprisingly, quantification of 5-HT+ fibers 317 318 ipsilateral to injury revealed enhanced serotonergic sprouting in apoE4 females compared to 319 apoE3 (Fig. 5 A-C Student's t-test, p=0.0193). This contradicted our expectation that a blunted 320 respiratory response to IH would correspond with attenuated fiber sprouting after injury.

321 Quantification of 5-HT fibers contralateral to injury showed no significant difference between E3

- 322 and E4 females, although E4 tended to have more 5-HT staining (Fig. 5 D, Student's t-test,
- 323 p=0.286). After injury, E4 females had more 5-HT fibers ipsilateral than contralateral to injury,
- 324 although this did not reach statistical significance (Fig. 5 *E*, Student's t-test, p=0.187).

### 325 Discussion

326 This study represents the first investigation into human genetic influences on the efficacy 327 of experimental therapeutic strategies for SCI. Our results demonstrate that individuals' 328 propensity for initiating beneficial neuroplastic responses to therapeutic IH is modified by sex 329 and apolipoprotein E genotype. By utilizing a well-described model of SCI and spinally-330 mediated motor plasticity, we provide evidence to support the hypothesis that human genetic 331 factors that are not represented by preclinical animal models limit the potential for recovery after 332 SCI. Our physiology and histology data indicate that sex and genotype influence the CNS 333 response to injury and therapeutic intervention, which poses a significant challenge to translating 334 one-size-fits-all treatment strategies.

### 335 APOE genotype and respiratory motor plasticity

Recovery of breathing function is a top priority for people living with cervical SCI (Anderson, 2004). Intermittent hypoxia has promising potential to enhance spinal plasticity for the restoration of a variety of motor behaviors, including breathing (Fuller et al., 2003; Lovett-Barr et al., 2012; Trumbower et al., 2012). Studies by Wadhwa et al. (2008) and Tester et al. (2014) in human participants have demonstrated that ventilatory LTF is expressed by both male and female subjects, even when living with a chronic SCI. However, to our knowledge, the interaction of sex and genetic factors remains unexplored in the LTF literature. Preclinical studies that have addressed the impact of sex on respiratory motor plasticity revealed that sex
hormone levels have significant ramifications for the potential to induce plasticity, likely due to
the interaction of sex hormones and the serotonergic system (Zabka et al., 2001a, 2001b, 2003).
Additionally, Baker-Herman et al. (2010) found that rat strains of different genetic backgrounds
vary in their responses to IH, which was associated with differences in the expression of 5-HT<sub>2A</sub>
receptors on PMNs.

349 To further address how genetic variability impacts spinal plasticity, we examined the 350 efficacy of IH for inducing LTF in targeted replacement mice expressing the human apoE  $\varepsilon$ 3, 351 and ɛ4 alleles. Since apoE first gained notoriety as a genetic marker for Alzheimer's Disease 352 (AD), an extensive body of literature has investigated impact of the apoE isoforms in the brain. 353 The  $\varepsilon 4$  allele increases the risk of developing AD and lowers the age of onset in a dose-354 dependent manner (Corder et al., 1993; Saunders et al., 1993). E4 carriers display mitochondrial 355 dysfunction, aggravated neuroinflammatory responses to CNS damage, loss of blood brain 356 barrier integrity, and impaired synaptic plasticity (Safieh et al., 2019). These factors are also key 357 determinants for the extent of tissue damage, plasticity, regeneration, and the potential for 358 recovery after SCI (Noble & Wrathall, 1989; P. G. Sullivan et al., 2007; Alilain & Goshgarian, 359 2008; Kigerl et al., 2009).

ApoE further became a gene of interest in our investigation after studies in human SCI
patients found that people who carried the ɛ4 allele experienced significantly less motor recovery
than non-carriers, despite spending more time in rehabilitation (Jha et al., 2008; Sun et al., 2011).
ApoE4 is known to curb recycling of NMDA and AMPA receptors to the postsynaptic
membrane and reduces levels of BDNF in the CNS (Chen et al., 2010; Chhibber & Zhao, 2017;
Sen et al., 2017). Since LTF requires BDNF signaling and activation of NMDA receptors,

individuals expressing the ɛ4 allele may have a constrained response to IH (Baker-Herman et al.,
2004; McGuire et al., 2005). However, our data demonstrates that mice expressing human apoE
isoforms did not differ in their diaphragmatic response to IH, indicating that there may be no
effect on LTF when *APOE* genotype is the sole variable being considered,

370 The lack of divergence between genotypes and the absence of augmented diaphragmatic 371 activity in response to IH could also be due to metabolic changes. Many protocols for the 372 induction of LTF, which are primarily conducted in rats, include the measurement of  $pCO_2$ 373 throughout recording (Hayashi et al., 1993; Bach & Mitchell, 1996). This measurement provides 374 a gauge of how metabolism is changing as a result of hypoxic hypometabolism: instead of 375 increasing respiratory activity, small mammals respond to hypoxic conditions by downregulating 376 their metabolic rate to reduce oxygen consumption (Hill, 1959). Because we did not measure 377  $pCO_2$  during EMG recordings due to the low blood volume of mice, we were unable to control 378 for changes in metabolic rate, which could have prevented IH-induced breathing augmentation. 379 However, we do not think that hypometabolism was responsible for masking genotype effects 380 since differences emerged when animals were grouped according to sex.

381 Interestingly, mice displayed a decrease in ipsilateral hemidiaphragmatic activity during 382 hypoxic bouts, instead of the heightened activity that is typical of the HVR observed in rats 383 (Pamenter & Powell, 2016). Very little data is available on the respiratory response to IH and 384 manifestation of LTF in C2 hemisected mice, although Minor et al. (2006) demonstrated the 385 presence and viability of the murine crossed phrenic pathway (CPP), the anatomical substrate 386 that mediates LTF (Golder and Mitchell, 2005). The few studies performed in mice are variable 387 in IH protocols and methods of assessing LTF (Terada et al., 2008; Hickner et al., 2014; 388 ElMallah et al., 2016). Our HVR data indicates that mice respond to bouts of hypoxia differently than rats, but further experiments are needed to characterize this phenomenon. Considering the
availability of transgenic mouse models, a standardized protocol for inducing and evaluating
LTF in murine models would be extremely advantageous for studying how human genes
influence spinally-mediated breathing plasticity.

393 Sex effects

394 Another explanation for the lack of observable differences between genotypes is that they 395 could be masked by sex effects. APOE has long been studied in the Alzheimer's field, where 396 genotype influences are known to be modulated by sex. While expression of the  $\varepsilon$ 4 allele 397 increases the risk of developing AD in both males and females, this risk is greater in females 398 (Duara et al., 1996; Altmann et al., 2014). In rodents, apoE4-related deficits in learning and 399 memory are aggravated in females, indicating that synaptic plasticity in the brain is impaired in a 400 sex-dependent manner (Raber et al., 1998; Kulkarni et al., 2020). The implication for similar 401 trends in spinally-mediated plasticity led to further analysis of our data, in which the influence of 402 genotype was investigated separately in males and females.

Diaphragmatic EMG recordings from females revealed a significant difference between the response of apoE3 and apoE4 animals. Forty minutes after IH, diaphragmatic activity was significantly depressed in females expressing  $\varepsilon$ 4. Consistent with findings in the brain, this demonstrates that apoE4 females have a limited propensity for plasticity in the spinal cord. This pattern was not reflected in male mice. In contrast with the current body of literature, we show that apoE3 males experience a barrier to synaptic plasticity, as they display the largest decrease in diaphragmatic activity.

410	To our knowledge, apoE3-associated attenuations of plasticity have never been reported
411	in young adult mice. Since the majority of apoE literature describes its effects on the brain, it is
412	possible that our results are due to a unique action of apoE in the spinal cord. The mechanism
413	behind induction of LTF in the bulbospinal breathing circuitry is similar to that of long term
414	potentiation (LTP) in the hippocampus: both rely on synaptic strengthening brought about by
415	activation of postsynaptic NMDA receptors and signaling through ERK (English & Sweatt,
416	1997; McGuire et al., 2005; Hoffman et al., 2012). Disparate results from a variety of studies in
417	targeted replacement mice suggest apoE4 can be detrimental or beneficial to LTP depending on
418	brain region, sex, and age (Levi et al., 2003; Kitamura et al., 2004; Trommer et al., 2004;
419	Korwek et al., 2009). Taking this into account, it is less surprising to see that apoE3 also has the
420	potential to augment or impede similar mechanisms of plasticity. This effect may also be
421	dependent on age and region of the CNS.
422	The inhibitory PNN and serotonergic presence after C2Hx in targeted replacement mice
423	Following SCI, there is a dramatic upregulation of inhibitory CSPGs at the site of injury
424	and in denervated targets (Bradbury et al., 2002; Massey et al., 2008; Alilain et al., 2011).
425	Indeed, after dorsal column transections, there is an upregulation of the CSPG-containing PNN
426	around sensory nuclei (Massey et al., 2008) and in previous studies utilizing lateral C2
427	hemisections, PMNs became further encased by CSPGs and the PNN (Alilain et al., 2011).
428	Despite the abundance of evidence implicating the importance of CSPGs in limiting plasticity,
429	regeneration, and recovery (reviewed by Tran et al., 2018), the influence of human genetics (and
430	APOE alleles) on PNN structure and neuronal sprouting in the injured spinal cord has never been
431	investigated. However, a study of human brains indicated that apoE4 augments expression of a

432

433 extensive staining of PNN observed in E4 mice (Conejero-Goldberg et al., 2015). 434 Indeed, our findings indicate that apoE4 females exhibit a greater density of the PNN in 435 the ventral horn region containing PMNs after injury. Although PMNS were not discreetly 436 labeled, upregulation of the PNN at the C4 level ipsilateral to injury suggests that deficits in 437 respiratory motor plasticity could be a consequence of the PNN's numerous influences on CNS 438 function and plasticity. Appearance of the PNNs containing CSPGs during development ends 439 critical periods in which experience-dependent plasticity shapes neural circuitry. Degradation of 440 CSPGs reopens this critical period and restores synaptic plasticity in the adult CNS (Pizzorusso 441 et al., 2002). Following lateral spinal hemisection, increasing densities of CSPG molecules 442 impede calcium diffusion and block action potential conduction in intact axons that are spared by 443 the injury (Hrabětová et al., 2009; Hunanyan et al., 2010). These molecules also create an 444 inhibitory microenvironment that prevents sprouting and regeneration of fibers in the injured 445 spinal cord, including 5-HT fibers that have the potential to enhance functional recovery after 446 experimental SCI (Alilain et al., 2011; Warren et al., 2018; Warren & Alilain, 2019). 447 Since serotonergic signaling at the level of PMNs is crucial to induction of LTF (Bach & 448 Mitchell, 1996), we quantified 5-HT staining around the putative PMN in spinal cords from E3 449 and E4 females. Density of 5-HT fibers was higher in E4 females both contralateral and 450 ipsilateral to injury, although this difference did not reach statistical significance on the 451 contralateral side. This indicates that compared to apoE3, apoE4 females may have greater 452 serotonergic innervation of the PMN in the absence of injury. However, additional studies are 453 needed to determine whether females expressing £4 have greater serotonergic innervation prior

CSPG known as brevican in the brain of Alzheimer's patients, which could explain the more

to injury, as well as after C2Hx. Indeed, if this pattern is consistent regardless of injury status,

increased 5-HT fiber density could represent a compensatory mechanism that maintains motor
neuron excitability in these animals while combatting the loss of synaptic integrity over time that
is observed in E4 animals (Klein et al., 2010). The observed attenuation of LTF after injury may
therefore be due to apoE4-dependent decreases in 5-HT receptor expression on PMNs, or a result
of alterations downstream of 5-HT receptor activation in the signaling pathways that are
necessary for the induction of LTF.

Although the higher density of PNN in E4 females is not associated with a decrease in the 461 462 amount of 5-HT at the level of the PMN after injury, the CSPG-containing PNN could still play a 463 role in abrogating respiratory motor plasticity. Further investigations are needed to determine 464 whether CSPGs block ion flow in spared axons such as the CPP after cervical hemisection 465 similar to the inhibition observed after thoracic injury (Hrabětová et al., 2009; Hunanyan et al., 466 2010). Previous studies have shown that degradation of CSPGs leads to increased presence of 5-467 HT around PMNs, which is associated with recovery of breathing function. However, these 468 studies did not address at the effect of CSPG upregulation or degradation on glutamatergic 469 sprouting or regeneration (Alilain et al., 2011; Warren et al., 2018; Warren & Alilain, 2019). 470 Therefore, alterations glutamatergic innervation of PMNs could also contribute to the 471 enhancement of diaphragmatic function demonstrated in these studies. Although E4 females 472 displayed more 5-HT fibers than E3 females, further examination of glutamatergic axon 473 regeneration and sprouting, as well as how enzymatic degradation of CSPGs alters this 474 innervation, could provide insight into whether PNN upregulation contributes to a lack of 475 respiratory motor plasticity in females expressing ɛ4.

The primary goal of this study was to investigate the role of genetic variability indetermining an individual's propensity for spinal plasticity and recovery of breathing function

478	after SCI. Preclinical studies typically test therapeutic approaches in a homogenous group of
479	animals, which does not represent the diversity found in the human population. As IH and other
480	therapeutics enter clinical trials, their efficacy for treating a heterogeneous population is an
481	important consideration. Overall, our findings that sex and APOE genotype modulate the
482	response to therapeutic IH, along with the current dearth of successful treatment strategies for
483	SCI, emphasizes the importance of advancing personalized medicine to improve outcomes for
484	injured individuals.

### 485 References

486	Ahuja, C. S., Nori, S., Tetreault, L., Wilson, J., Kwon, B., Harrop, J., Choi, D., & Fehlings, M. G.
487	(2017). Traumatic Spinal Cord Injury—Repair and Regeneration. Neurosurgery, 80(3S),
488	S9–S22. https://doi.org/10.1093/neuros/nyw080
489	Alilain, W. J., & Goshgarian, H. G. (2008). Glutamate receptor plasticity and activity-regulated
490	cytoskeletal associated protein regulation in the phrenic motor nucleus may mediate
491	spontaneous recovery of the hemidiaphragm following chronic cervical spinal cord injury.
492	Experimental Neurology, 212(2), 348–357.
493	https://doi.org/10.1016/j.expneurol.2008.04.017
494	Alilain, W. J., Horn, K. P., Hu, H., Dick, T. E., & Silver, J. (2011). Functional regeneration of
495	respiratory pathways after spinal cord injury. Nature, 475(7355), 196–200.
496	https://doi.org/10.1038/nature10199
497	Alp, E., & Voss, A. (2006). Ventilator associated pneumonia and infection control. Annals of
498	Clinical Microbiology and Antimicrobials, 5, 7. https://doi.org/10.1186/1476-0711-5-7
499	Altmann, A., Tian, L., Henderson, V. W., Greicius, M. D., & Alzheimer's Disease Neuroimaging
500	Initiative Investigators. (2014). Sex modifies the APOE-related risk of developing
501	Alzheimer disease. Annals of Neurology, 75(4), 563-573.
502	https://doi.org/10.1002/ana.24135
503	Anderson, K. D. (2004). Targeting recovery: Priorities of the spinal cord-injured population.
504	Journal of Neurotrauma, 21(10), 1371–1383. https://doi.org/10.1089/neu.2004.21.1371
505	Bach, K. B., & Mitchell, G. S. (1996). Hypoxia-induced long-term facilitation of respiratory
506	activity is serotonin dependent. Respiration Physiology, 104(2-3), 251-260.
507	Baker, T. L., & Mitchell, G. S. (2000). Episodic but not continuous hypoxia elicits long-term
508	facilitation of phrenic motor output in rats. The Journal of Physiology, 529(1), 215–219.
509	https://doi.org/10.1111/j.1469-7793.2000.00215.x

510	Baker-Herman, T.L., Bavis, R. W., Dahlberg, J. M., Mitchell, A. Z., Wilkerson, J. E. R., Golder,
511	F. J., MacFarlane, P. M., Watters, J. J., Behan, M., & Mitchell, G. S. (2010). Differential
512	expression of respiratory long-term facilitation among inbred rat strains. Respiratory
513	Physiology & Neurobiology, 170(3), 260–267. https://doi.org/10.1016/j.resp.2009.12.008
514	Baker-Herman, Tracy L., Fuller, D. D., Bavis, R. W., Zabka, A. G., Golder, F. J., Doperalski, N.
515	J., Johnson, R. A., Watters, J. J., & Mitchell, G. S. (2004). BDNF is necessary and
516	sufficient for spinal respiratory plasticity following intermittent hypoxia. Nature
517	Neuroscience, 7(1), 48–55. https://doi.org/10.1038/nn1166
518	Bergofsky, E. H. (1964). Mechanism for respiratory insufficiency after cervical cord injury; a
519	source of alveolar hypoventilation. Annals of Internal Medicine, 61, 435-447.
520	https://doi.org/10.7326/0003-4819-61-3-435
521	Boyles, J. K., Pitas, R. E., Wilson, E., Mahley, R. W., & Taylor, J. M. (1985, October 1).
522	Apolipoprotein E associated with astrocytic glia of the central nervous system and with
523	nonmyelinating glia of the peripheral nervous system. https://doi.org/10.1172/JCI112130
524	Bradbury, E. J., Moon, L. D. F., Popat, R. J., King, V. R., Bennett, G. S., Patel, P. N., Fawcett, J.
525	W., & McMahon, S. B. (2002). Chondroitinase ABC promotes functional recovery after
526	spinal cord injury. <i>Nature</i> , <i>416</i> (6881), 636–640. https://doi.org/10.1038/416636a
527	Chen, M. S., Huber, A. B., van der Haar, M. E., Frank, M., Schnell, L., Spillmann, A. A., Christ,
528	F., & Schwab, M. E. (2000). Nogo-A is a myelin-associated neurite outgrowth inhibitor
529	and an antigen for monoclonal antibody IN-1. Nature, 403(6768), 434–439.
530	https://doi.org/10.1038/35000219
531	Chen, Y., Durakoglugil, M. S., Xian, X., & Herz, J. (2010). ApoE4 reduces glutamate receptor
532	function and synaptic plasticity by selectively impairing ApoE receptor recycling.
533	Proceedings of the National Academy of Sciences of the United States of America,
534	107(26), 12011–12016. https://doi.org/10.1073/pnas.0914984107

535	Chhibber, A., & Zhao, L. (2017). ER $\beta$ and ApoE isoforms interact to regulate BDNF-5-HT2A
536	signaling and synaptic function in the female brain. Alzheimer's Research & Therapy,
537	9(1), 79. https://doi.org/10.1186/s13195-017-0305-3
538	Conejero-Goldberg, C., Hyde, T. M., Chen, S., Herman, M. M., Kleinman, J. E., Davies, P., &
539	Goldberg, T. E. (2015). Cortical Transcriptional Profiles in APOE4 Carriers with
540	Alzheimer's Disease: Patterns of Protection and Degeneration. Journal of Alzheimer's
541	<i>Disease</i> , <i>48</i> (4), 969–978. https://doi.org/10.3233/JAD-150345
542	Corder, E. H., Saunders, A. M., Strittmatter, W. J., Schmechel, D. E., Gaskell, P. C., Small, G.
543	W., Roses, A. D., Haines, J. L., & Pericak-Vance, M. A. (1993). Gene dose of
544	apolipoprotein E type 4 allele and the risk of Alzheimer's disease in late onset families.
545	Science (New York, N.Y.), 261(5123), 921–923. https://doi.org/10.1126/science.8346443
546	David, S., & Aguayo, A. J. (1981). Axonal elongation into peripheral nervous system "bridges"
547	after central nervous system injury in adult rats. Science (New York, N.Y.), 214(4523),
548	931–933. https://doi.org/10.1126/science.6171034
549	DeVivo, M. J., & Ivie, C. S. (1995). Life expectancy of ventilator-dependent persons with spinal
550	cord injuries. <i>Chest</i> , <i>108</i> (1), 226–232.
551	Duara, R., Barker, W. W., Lopez-Alberola, R., Loewenstein, D. A., Grau, L. B., Gilchrist, D.,
552	Sevush, S., & St George-Hyslop, S. (1996). Alzheimer's disease: Interaction of
553	apolipoprotein E genotype, family history of dementia, gender, education, ethnicity, and
554	age of onset. Neurology, 46(6), 1575–1579. https://doi.org/10.1212/wnl.46.6.1575
555	ElMallah, M. K., Stanley, D. A., Lee, KZ., Turner, S. M. F., Streeter, K. A., Baekey, D. M., &
556	Fuller, D. D. (2016). Power spectral analysis of hypoglossal nerve activity during
557	intermittent hypoxia-induced long-term facilitation in mice. Journal of Neurophysiology,
558	115(3), 1372–1380. https://doi.org/10.1152/jn.00479.2015

English, J. D., & Sweatt, J. D. (1997). A requirement for the mitogen-activated protein kinase
cascade in hippocampal long term potentiation. The Journal of Biological Chemistry,
272(31), 19103–19106. https://doi.org/10.1074/jbc.272.31.19103
Fouad, K., Popovich, P. G., Kopp, M. A., & Schwab, J. M. (2020). The neuroanatomical-
functional paradox in spinal cord injury. Nature Reviews. Neurology.
https://doi.org/10.1038/s41582-020-00436-x
Fuller, D. D, Bach, K. B., Baker, T. L., Kinkead, R., & Mitchell, G. S. (2000). Long term
facilitation of phrenic motor output. Respiration Physiology, 121(2), 135–146.
https://doi.org/10.1016/S0034-5687(00)00124-9
Fuller, David D., Johnson, S. M., Olson, E. B., & Mitchell, G. S. (2003). Synaptic pathways to
phrenic motoneurons are enhanced by chronic intermittent hypoxia after cervical spinal
cord injury. The Journal of Neuroscience: The Official Journal of the Society for
Neuroscience, 23(7), 2993–3000.
Golder, F. J. (2005). Spinal Synaptic Enhancement with Acute Intermittent Hypoxia Improves
Respiratory Function after Chronic Cervical Spinal Cord Injury. Journal of Neuroscience,
25(11), 2925–2932. https://doi.org/10.1523/JNEUROSCI.0148-05.2005
Hayashi, F., Coles, S. K., Bach, K. B., Mitchell, G. S., & McCrimmon, D. R. (1993). Time-
dependent phrenic nerve responses to carotid afferent activation: Intact vs.
decerebellate rats. The American Journal of Physiology, 265(4 Pt 2), R811-819.
https://doi.org/10.1152/ajpregu.1993.265.4.R811
Hayes, H. B., Jayaraman, A., Herrmann, M., Mitchell, G. S., Rymer, W. Z., & Trumbower, R. D.
(2014). Daily intermittent hypoxia enhances walking after chronic spinal cord injury: A
randomized trial. Neurology, 82(2), 104–113.
https://doi.org/10.1212/01.WNL.0000437416.34298.43
Hickner, S., Hussain, N., Angoa-Perez, M., Francescutti, D. M., Kuhn, D. M., & Mateika, J. H.

(2014). Ventilatory long-term facilitation is evident after initial and repeated exposure to

585	intermittent hypoxia in mice genetically depleted of brain serotonin. Journal of Applied	
586	Physiology (Bethesda, Md. : 1985), 116(3), 240–250.	
587	https://doi.org/10.1152/japplphysiol.01197.2013	
588	Hill, J. R. (1959). The oxygen consumption of new-born and adult mammals. Its dependence on	
589	the oxygen tension in the inspired air and on the environmental temperature. The	
590	Journal of Physiology, 149(2), 346–373.	
591	Hoffman, M. S., Nichols, N. L., Macfarlane, P. M., & Mitchell, G. S. (2012). Phrenic long-term	
592	facilitation after acute intermittent hypoxia requires spinal ERK activation but not TrkB	
593	synthesis. Journal of Applied Physiology, 113(8), 1184–1193.	
594	https://doi.org/10.1152/japplphysiol.00098.2012	
595	Hrabětová, S., Masri, D., Tao, L., Xiao, F., & Nicholson, C. (2009). Calcium diffusion enhanced	
596	after cleavage of negatively charged components of brain extracellular matrix by	
597	chondroitinase ABC. The Journal of Physiology, 587(Pt 16), 4029–4049.	
598	https://doi.org/10.1113/jphysiol.2009.170092	
599	Huie, J. R., Morioka, K., Haefeli, J., & Ferguson, A. R. (2017). What Is Being Trained? How	
600	Divergent Forms of Plasticity Compete To Shape Locomotor Recovery after Spinal Cord	
601	Injury. Journal of Neurotrauma, 34(10), 1831–1840.	
602	https://doi.org/10.1089/neu.2016.4562	
603	Hunanyan, A. S., García-Alías, G., Alessi, V., Levine, J. M., Fawcett, J. W., Mendell, L. M., &	
604	Arvanian, V. L. (2010). Role of Chondroitin Sulfate Proteoglycans in Axonal Conduction	
605	in Mammalian Spinal Cord. The Journal of Neuroscience, 30(23), 7761–7769.	
606	https://doi.org/10.1523/JNEUROSCI.4659-09.2010	
607	Jack, A. S., Hurd, C., Martin, J., & Fouad, K. (2020). Electrical Stimulation as a Tool to Promote	
608	Plasticity of the Injured Spinal Cord. Journal of Neurotrauma, 37(18), 1933–1953.	
609	https://doi.org/10.1089/neu.2020.7033	

610	Jha, A., Lammertse, D. P., Coll, J. R., Charlifue, S., Coughlin, C. T., Whiteneck, G. G., &
611	Worley, G. (2008). Apolipoprotein E epsilon4 allele and outcomes of traumatic spinal
612	cord injury. The Journal of Spinal Cord Medicine, 31(2), 171–176.
613	Kigerl, K. A., Gensel, J. C., Ankeny, D. P., Alexander, J. K., Donnelly, D. J., & Popovich, P. G.
614	(2009). Identification of Two Distinct Macrophage Subsets with Divergent Effects
615	Causing either Neurotoxicity or Regeneration in the Injured Mouse Spinal Cord. The
616	Journal of Neuroscience, 29(43), 13435–13444.
617	https://doi.org/10.1523/JNEUROSCI.3257-09.2009
618	Kitamura, H. W., Hamanaka, H., Watanabe, M., Wada, K., Yamazaki, C., Fujita, S. C., Manabe,
619	T., & Nukina, N. (2004). Age-dependent enhancement of hippocampal long-term
620	potentiation in knock-in mice expressing human apolipoprotein E4 instead of mouse
621	apolipoprotein E. Neuroscience Letters, 369(3), 173–178.
622	https://doi.org/10.1016/j.neulet.2004.07.084
623	Klein, R. C., Mace, B. E., Moore, S. D., & Sullivan, P. M. (2010). Progressive loss of synaptic
624	integrity in human apolipoprotein E4 targeted replacement mice and attenuation by
625	apolipoprotein E2. Neuroscience, 171(4), 1265–1272.
626	https://doi.org/10.1016/j.neuroscience.2010.10.027
627	Knouff, C., Hinsdale, M. E., Mezdour, H., Altenburg, M. K., Watanabe, M., Quarfordt, S. H.,
628	Sullivan, P. M., & Maeda, N. (1999). Apo E structure determines VLDL clearance and
629	atherosclerosis risk in mice. The Journal of Clinical Investigation, 103(11), 1579–1586.
630	https://doi.org/10.1172/JCI6172
631	Korwek, K. M., Trotter, J. H., LaDu, M. J., Sullivan, P. M., & Weeber, E. J. (2009). ApoE
632	isoform-dependent changes in hippocampal synaptic function. Molecular
633	Neurodegeneration, 4, 21. https://doi.org/10.1186/1750-1326-4-21
634	Kulkarni, P., Grant, S., Morrison, T. R., Cai, X., Iriah, S., Kristal, B. S., Honeycutt, J., Brenhouse,
635	H., Hartner, J. C., Madularu, D., & Ferris, C. F. (2020). Characterizing the human APOE

636	epsilon 4 knock-in transgene in female and male rats with multimodal magnetic	
637	resonance imaging. Brain Research, 1747, 147030.	
638	https://doi.org/10.1016/j.brainres.2020.147030	
639	Levi, O., Jongen-Relo, A. L., Feldon, J., Roses, A. D., & Michaelson, D. M. (2003). ApoE4	
640	impairs hippocampal plasticity isoform-specifically and blocks the environmental	
641	stimulation of synaptogenesis and memory. Neurobiology of Disease, 13(3), 273–282.	
642	https://doi.org/10.1016/S0969-9961(03)00045-7	
643	Lovett-Barr, M. R., Satriotomo, I., Muir, G. D., Wilkerson, J. E. R., Hoffman, M. S., Vinit, S., &	
644	Mitchell, G. S. (2012). Repetitive intermittent hypoxia induces respiratory and somatic	
645	motor recovery after chronic cervical spinal injury. The Journal of Neuroscience: The	
646	Official Journal of the Society for Neuroscience, 32(11), 3591–3600.	
647	https://doi.org/10.1523/JNEUROSCI.2908-11.2012	
648	Lynch, M., Duffell, L., Sandhu, M., Srivatsan, S., Deatsch, K., Kessler, A., Mitchell, G. S.,	
649	Jayaraman, A., & Rymer, W. Z. (2017). Effect of acute intermittent hypoxia on motor	
650	function in individuals with chronic spinal cord injury following ibuprofen pretreatment: A	
651	pilot study. The Journal of Spinal Cord Medicine, 40(3), 295–303.	
652	https://doi.org/10.1080/10790268.2016.1142137	
653	Mahley, R. W. (2016). Apolipoprotein E: From cardiovascular disease to neurodegenerative	
654	disorders. Journal of Molecular Medicine (Berlin, Germany), 94, 739–746.	
655	https://doi.org/10.1007/s00109-016-1427-y	
656	Main, B. S., Villapol, S., Sloley, S. S., Barton, D. J., Parsadanian, M., Agbaegbu, C., Stefos, K.,	
657	McCann, M. S., Washington, P. M., Rodriguez, O. C., & Burns, M. P. (2018).	
658	Apolipoprotein E4 impairs spontaneous blood brain barrier repair following traumatic	
659	brain injury. Molecular Neurodegeneration, 13(1), 17. https://doi.org/10.1186/s13024-	
660	018-0249-5	

661	Massey, J. M., Amps, J., Viapiano, M. S., Matthews, Russell. T., Wagoner, M. R., Whitaker, C.
662	M., Alilain, W., Yonkof, A. L., Khalyfa, A., Cooper, N. G. F., Silver, J., & Onifer, S. M.
663	(2008). Increased Chondroitin Sulfate Proteoglycan Expression in Denervated Brainstem
664	Targets Following Spinal Cord Injury Creates a Barrier to Axonal Regeneration
665	Overcome by Chondroitinase ABC and Neurotrophin-3. Experimental Neurology, 209(2),
666	426–445. https://doi.org/10.1016/j.expneurol.2007.03.029
667	McGuire, M., Zhang, Y., White, D. P., & Ling, L. (2005). Phrenic long-term facilitation requires
668	NMDA receptors in the phrenic motonucleus in rats. The Journal of Physiology, 567(Pt
669	2), 599–611. https://doi.org/10.1113/jphysiol.2005.087650
670	Minor, K. H., Akison, L. K., Goshgarian, H. G., & Seeds, N. W. (2006). Spinal cord injury-
671	induced plasticity in the mouse—The crossed phrenic phenomenon. Experimental
672	Neurology, 200(2), 486–495. https://doi.org/10.1016/j.expneurol.2006.02.125
673	National Spinal Cord Injury Statistical Center. (2018). Spinal cord Injury Facts and Figures at a
674	Glance. https://www.nscisc.uab.edu/
675	Navarrete-Opazo, A., & Mitchell, G. S. (2014). Therapeutic potential of intermittent hypoxia: A
676	matter of dose. American Journal of Physiology - Regulatory, Integrative and
677	Comparative Physiology, 307(10), R1181–R1197.
678	https://doi.org/10.1152/ajpregu.00208.2014
679	Noble, L. J., & Wrathall, J. R. (1989). Distribution and time course of protein extravasation in the
680	rat spinal cord after contusive injury. Brain Research, 482(1), 57–66.
681	https://doi.org/10.1016/0006-8993(89)90542-8
682	Pamenter, M. E., & Powell, F. L. (2016). Time Domains of the Hypoxic Ventilatory Response
683	and Their Molecular Basis. Comprehensive Physiology, 6(3), 1345–1385.
684	https://doi.org/10.1002/cphy.c150026
685	Park, K. K., Liu, K., Hu, Y., Smith, P. D., Wang, C., Cai, B., Xu, B., Connolly, L., Kramvis, I.,

686 Sahin, M., & He, Z. (2008). Promoting Axon Regeneration in the Adult CNS by



713 replacement of mouse Apoe with human Apoe\*2. The Journal of Clinical Investigation, 714 102(1), 130-135. https://doi.org/10.1172/JCI2673 Sullivan, Patrick M., Mezdour, H., Aratani, Y., Knouff, C., Najib, J., Reddick, R. L., Quarfordt, S. 715 716 H., & Maeda, N. (1997). Targeted Replacement of the Mouse Apolipoprotein E Gene 717 with the Common Human APOE3 Allele Enhances Diet-induced Hypercholesterolemia 718 and Atherosclerosis. Journal of Biological Chemistry, 272(29), 17972-17980. 719 https://doi.org/10.1074/jbc.272.29.17972 720 Sun, C., Ji, G., Liu, Q., & Yao, M. (2011). Apolipoprotein E epsilon 4 allele and outcomes of 721 traumatic spinal cord injury in a Chinese Han population. Molecular Biology Reports, 722 38(7), 4793-4796. https://doi.org/10.1007/s11033-010-0620-2 723 Terada, J., Nakamura, A., Zhang, W., Yanagisawa, M., Kuriyama, T., Fukuda, Y., & Kuwaki, T. 724 (2008). Ventilatory long-term facilitation in mice can be observed during both sleep and 725 wake periods and depends on orexin. Journal of Applied Physiology, 104(2), 499-507. 726 https://doi.org/10.1152/japplphysiol.00919.2007 727 Tester, N. J., Fuller, D. D., Fromm, J. S., Spiess, M. R., Behrman, A. L., & Mateika, J. H. (2014). 728 Long-term facilitation of ventilation in humans with chronic spinal cord injury. American 729 Journal of Respiratory and Critical Care Medicine, 189(1), 57–65. 730 https://doi.org/10.1164/rccm.201305-0848OC 731 Tom, V. J., Kadakia, R., Santi, L., & Houlé, J. D. (2009). Administration of chondroitinase ABC 732 rostral or caudal to a spinal cord injury site promotes anatomical but not functional 733 plasticity. Journal of Neurotrauma, 26(12), 2323-2333. 734 https://doi.org/10.1089/neu.2009.1047 735 Tran, A. P., Warren, P. M., & Silver, J. (2018). The Biology of Regeneration Failure and 736 Success After Spinal Cord Injury. Physiological Reviews, 98(2), 881–917. 737 https://doi.org/10.1152/physrev.00017.2017

738	Trommer, B. L., Shah, C., Yun, S. H., Gamkrelidze, G., Pasternak, E. S., Ye, G. L., Sotak, M.,
739	Sullivan, P. M., Pasternak, J. F., & LaDu, M. J. (2004). ApoE isoform affects LTP in
740	human targeted replacement mice. Neuroreport, 15(17), 2655–2658.
741	https://doi.org/10.1097/00001756-200412030-00020
742	Trumbower, R. D., Hayes, H. B., Mitchell, G. S., Wolf, S. L., & Stahl, V. A. (2017). Effects of
743	acute intermittent hypoxia on hand use after spinal cord trauma: A preliminary study.
744	Neurology, 89(18), 1904–1907. https://doi.org/10.1212/WNL.000000000004596
745	Trumbower, R. D., Jayaraman, A., Mitchell, G. S., & Rymer, W. Z. (2012). Exposure to Acute
746	Intermittent Hypoxia Augments Somatic Motor Function in Humans With Incomplete
747	Spinal Cord Injury. Neurorehabilitation and Neural Repair, 26(2), 163–172.
748	https://doi.org/10.1177/1545968311412055
749	Urban, M. W., Ghosh, B., Block, C. G., Strojny, L. R., Charsar, B. A., Goulão, M., Komaravolu,
750	S. S., Smith, G. M., Wright, M. C., Li, S., & Lepore, A. C. (2019). Long-Distance Axon
751	Regeneration Promotes Recovery of Diaphragmatic Respiratory Function after Spinal
752	Cord Injury. ENeuro, 6(5). https://doi.org/10.1523/ENEURO.0096-19.2019
753	Wadhwa, H., Gradinaru, C., Gates, G. J., Badr, M. S., & Mateika, J. H. (2008). Impact of
754	intermittent hypoxia on long-term facilitation of minute ventilation and heart rate
755	variability in men and women: Do sex differences exist? Journal of Applied Physiology
756	(Bethesda, Md.: 1985), 104(6), 1625–1633.
757	https://doi.org/10.1152/japplphysiol.01273.2007
758	Warren, Philippa M., Steiger, S. C., Dick, T. E., MacFarlane, P. M., Alilain, W. J., & Silver, J.
759	(2018). Rapid and robust restoration of breathing long after spinal cord injury. Nature
760	Communications, 9(1), 4843. https://doi.org/10.1038/s41467-018-06937-0
761	Warren, Philippa Mary, & Alilain, W. J. (2019). Plasticity Induced Recovery of Breathing Occurs
762	at Chronic Stages after Cervical Contusion. Journal of Neurotrauma, 36(12), 1985–1999.

763 https://doi.org/10.1089/neu.2018.6186

764	Zabka, A. G., Behan, M., & Mitchell, G. S. (2001a). Long term facilitation of respiratory motor
765	output decreases with age in male rats. The Journal of Physiology, 531(Pt 2), 509–514.
766	https://doi.org/10.1111/j.1469-7793.2001.0509i.x
767	Zabka, A. G., Behan, M., & Mitchell, G. S. (2001b). Selected contribution: Time-dependent
768	hypoxic respiratory responses in female rats are influenced by age and by the estrus
769	cycle. Journal of Applied Physiology (Bethesda, Md.: 1985), 91(6), 2831–2838.
770	https://doi.org/10.1152/jappl.2001.91.6.2831
771	Zabka, A. G., Mitchell, G. S., Olson, E. B., & Behan, M. (2003). Selected contribution: Chronic
772	intermittent hypoxia enhances respiratory long-term facilitation in geriatric female rats.
773	Journal of Applied Physiology (Bethesda, Md.: 1985), 95(6), 2614–2623; discussion
774	2604. https://doi.org/10.1152/japplphysiol.00476.2003
775	Zhao, N., Liu, CC., Qiao, W., & Bu, G. (2018). Apolipoprotein E, Receptors and Modulation of
776	Alzheimer's Disease. Biological Psychiatry, 83(4), 347–357.
777	https://doi.org/10.1016/j.biopsych.2017.03.003
778	Zholudeva, L. V., Qiang, L., Marchenko, V., Dougherty, K. J., Sakiyama-Elbert, S. E., & Lane,
779	M. A. (2018). The Neuroplastic and Therapeutic Potential of Spinal Interneurons in the
780	Injured Spinal Cord. Trends in Neurosciences, 41(9), 625–639.
781	https://doi.org/10.1016/j.tins.2018.06.004
782	Zhou, W., Xu, D., Peng, X., Zhang, Q., Jia, J., & Crutcher, K. A. (2008). Meta-analysis of
783	APOE4 allele and outcome after traumatic brain injury. Journal of Neurotrauma, 25(4),
784	279–290. https://doi.org/10.1089/neu.2007.0489

### 786 Legends

787 Table 1.

788 Group sizes for diaphragmatic EMGs

789 Figure 1.

790 Magnitude of respiratory motor plasticity is not determined by apoE genotype alone. A.

- 791 Timeline of intermittent hypoxia protocol. Green arrows represent time point at which peak
- amplitude was analyzed. B. Schematic of the neural circuitry that mediates breathing. Location
- 793 of the left C2 hemisection is indicated by red X. C. Representative image of cresyl violet staining

794 of the spinal cord at the C2 level after injury. D. Quantification of diaphragmatic amplitude over

- time during and after IH. There was no difference between genotypes in the change in amplitude
- 796 over time (a. RM ANOVA p=0.741, MD=0.043, CI=-0.22 to 0.31) E. Quantification of
- 797 diaphragmatic amplitude during the first normoxic bout and 40 minutes after IH (b. Paired t-test
- 798 E3 normoxia v. 40min p=0.741, MD=0.029, CI=-0.16 to 0.22, c. E4 normoxia v. 40min p=0.405,

MD=0.084, CI=-0.29 to 0.12). Bars show mean and SEM values.

800 Figure 2.

- 801 ApoE3 males demonstrate a trend of decreasing diaphragmatic activity in response to IH.
- 802 A. Representative traces of diaphragmatic activity during the first normoxic bout and 40 minutes

803 after IH. B. Representative traces from male mice that had no spontaneous recovery. C.

- 804 Quantification of diaphragmatic activity over time during and after IH. Amplitude of
- 805 diaphragmatic bursts is not significantly different between E3 and E4 animals. D. Quantification
- 806 of diaphragmatic activity during the first normoxic bout and 40 minutes after IH (d. RM
- 807 ANOVA E3/E3 t=0.03, MD=0.0086, CI=-0.295 to 0.36 e. E3/E4 norm t=-0.55, MD=0.23, CI=-

808 1.18 to 0.078, f. E3/E4 40 min t=-0.91, MD=0.35, CI=-1.54 to -0.28 g. E4/E4 t=-0.67,

809 MD=0.11, CI=-0.41 to 0.63). Bars show mean and SEM values.

810 Figure 3.

### 811 ApoE4 females display significantly less diaphragmatic activity than E3 females after IH.

812 A. representative traces of diaphragmatic activity during the first normoxic bout and 40 minutes

813 after IH. B. Representative traces from an apoE4 female non-responder. C. Quantification of

- 814 diaphragmatic activity over time during and after IH. Amplitude of diaphragmatic bursts are
- 815 significantly greater in E3 females than in E4. D. Quantification of diaphragmatic activity during
- the first normoxic bout and 40 minutes after IH (h. RM ANOVA E3/E3 t=-0.32, MD=0.071,

817 CI=-0.75to 0.11, *i*. E3/E4 normoxia t=0.84, MD=0.18, CI=0.43 to 1.25, *j*. E3/E4 40 min t=2.08,

818 MD=0.44, CI=1.67 to 2.49, *k*. E4/E4 t=1.01, MD=0.19, CI=0.64 to 1.38). Bars show mean and

819 SEM values.

820 Figure 4.

E4 females have higher levels of PNN. *A.* Representative images of WFA staining at the C4
spinal cord level (DAPI is in red, WFA is in green). *B.* Higher magnification images show the
PNN surrounding putative phrenic motor neurons. *C.* Quantification of WFA indicates that
apoE4 mice express more WFA than E3 mice, although this trend is not statistically significant
(*l.* Welch's t-test p=0.0697, MD=0.0029, CI=-0.00036 to 0.0061). Bars represent mean ± SEM.

826 Figure 5.

E4 females have higher density of spinal 5-HT fibers. *A*. Representative images of stained 5HT fibers in the C4 spinal cord level. Higher magnification shows individual 5-HT fibers in the
area of the putative PMN. *C*. Significantly more serotonergic fibers are found contralateral to

- 830 injury in apoE4 females (*m*. Student's t-test p=0.0193, MD=0.0016, CI=0.00036 to 0.0029). *D*.
- 5-HT fibers contralateral to injury at the C4 level (n. Student's t-test p=0.286, MD=0.00104,
- 832 CI=-0.0011 to 0.0032). E. Ipsilateral 5-HT staining normalized to contralateral (o. Student's t-
- test p=0.187, MD=0.47, CI=-0.29 to 1.24). Bars represent mean  $\pm$  SEM.

### 835 Extended Data 2-1.

### 836 The respiratory response to hypoxia is determined by APOE genotype in male mice. A.

- 837 Quantification of diaphragmatic burst frequency in male mice. There is no significant difference
- 838 between the decreases in apoE3 and apoE4 mice (RM ANOVA p=0.846). B, C Quantification of
- 839 diaphragmatic burst amplitude (B.) and frequency (C.) in response to a 10-minute hypoxic
- 840 exposure. Hypoxia appears to attenuate breathing in apoE3 males. No statistics were performed
- 841 due to low n. E3 and E4 n=2.

### 842 Extended Data 3-1.

### 843 Hypoxia induces a decline in breathing frequency in female APOE targeted replacement

- 844 **mice.** A. Quantification of diaphragmatic burst frequency in female mice. There is no significant
- 845 difference between the decreases in apoE3 and apoE4 mice (RM ANOVA p=0.673). B,C.
- 846 Quantification of diaphragmatic burst amplitude (B.) and frequency (C.) in response to a 10-
- 847 minute hypoxic exposure. Breathing frequency displayed a negative trend in apoE4 females. No
- statistics were performed due to low n. E3 n=3, E4 n=2.

### 850 Table 1.

	Male	Female
E3	n=7	n=8
E4	n=6	n=11











Α.